Percolation Model for Universal Scaling of Vegetation Net Primary Productivity with Climate Variables

Allen G. Hunt¹, Muhammad Sahimi², Behzad Ghanbarian³, and Germán Poveda⁴ ¹Department of Physics, Wright State University, Dayton, OH 45435, ²Mork Family Department of Chemical Engineering and Materials Science, University of Southern California, Los Angeles, CA 90089, ³Porous Media Research Lab, Department of Geology, Kansas State University, Manhattan, KS 66506, ⁴Department of Geosciences and Environment, Universidad Nacional de Colombia, Medellín, Colombia Abstract We show that a quantitative, physically-based model of water balance, combined with a model for the dependence of net primary productivity (NPP) of vegetation on evapotranspiration E leads to the universal scaling of NPP with the climate variables, namely, precipitation P and potential evapotranspiration E_p . The model for NPP is based on percolation theory. The water balance is derived based on the hypothesis that ecosystem selection is driven by optimizing NPP with respect to the principal hydrologic fluxes i.e., E and Q (=P-E; known as run-off), with the flux dependence of E being applied self-consistently to the NPP's model. The predictions are compared separately with Budyko's compilation of the NPP data as a function of E_p and P using the aridity index, $A_I = E_p/P$, and E_p for each set of experimental data, and are shown to be in excellent agreement with the data.

Plain Language Summary Predicting the dependence of the net primary production on the climate variables, namely, precipitation P and potential evapotranspiration E_p , has been the subject of numerous studies. We develop a model by combining percolation theory, which describes the morphology of heterogeneous soil and plants' root structure, with a model of water balance, which is based on the prevalent view in ecology that, to access and utilize resources for growth and reproduction, plant ecosystems exploit diversity optimally. We show that one obtains an equation describing universal scaling of NPP with P and E_p . The agreement between the predictions of the scaling equation and the relevant data at global scale is excellent.

1. Introduction

Earth's soil has been called "the crucible of life, a self-regulating, biophysical factory, acting like a composite living entity." (Hillel, 2005). It represents a critical nexus (Brantley et al., 2007; Anderson et al., 2007) where plants grow, carbon is sequestered, and precipitation Pis partitioned into run-off Q and evapotranspiration E (Falkenmark and Rockström, 2006; Gentine et al., 2012; Maxwell and Condon, 2016). Partitioning of the precipitated water into its primary fluxes has a profound effect on what happens in soil and the environment and, therefore, life and its diversity. Consider, for example, biodiversity. Precipitation promotes biodiversity (Krishnaswamy et al., 2009), which in turn promotes the assimilation of carbon by plants, i.e, the net primary productivity (NPP), N_p (Oehri et al., 2017). Despite its high importance to the food-energy-water nexus (D'Odorico et al., 2018) to understanding climate stability and change (Tapley et al., 2019) and for managing water resources and optimizing agricultural production (Huang et al., 2012), there is currently no parsimonious model for accurately predicting the NPP of individual ecosystems, as well as trends across biomes. Ever since the work of Rosenzweig (1968) and Webb et al. (1986), research on NPP has tended to focus on far more complex process-based models (Cramer et al., 1999; Adam et al., 2004). For Example, Raich et al. (1991) stated that as a preferred alternative, "We can more successfully predict rates of NPP in terrestrial ecosystems, if we model the basic processes controlling productivity and how they are influenced by environmental factors."

Nevertheless, it remains true that evapotranspiration E is the most important single input for predicting NPP (Rosenzweig, 1968), since it represents a measure of two highly variable inputs in photosynthesis, namely, light and water. The monotonic increase in NPP as a power of E(Rosenzweig, 1968) is bounded only by limits on solar energy (de Wit, 1965; Krause-Jensen and Sand-Jensen, 1998) at about 3500 g cm⁻² yr⁻¹ (Hunt, 2017). Predicting E as a function of climatological variables P and potential evapotranspiration E_p - the central problem of hydrology (Horton, 1931) and known as the water balance - is thus the key missing ingradient for a model of NPP in terms of climatological variables.

As a report by the National Academy of Sciences (1991) suggested, the foundation for a solution of the water balance must be based on the link between hydrology and ecological optimality, since the fluxes of the key biological variables, such as biomass, are also linked with such operating chemical processes as carbon assimilation, which in turn are linked to hydrological processes, such as evapotranspiration. Such relationships make it clear that hydrology is a fundamental component of the biogeochemical cycles. A recent theoretical development (Hunt et al., 2021) that accurately predicts E as a function of solar energy and precipitation, without using adjustable parameters, is in the spirit of such a strategy, which was derived by optimizing NPP with respect to the principal hydrologic fluxes.

In this Letter we propose to use the model by Hunt et al. (2021) for the evapotranspiration in order to develop a theoretical framework for predicting NPP as a function of solar energy and precipitation with *at most* one adjustable parameter. Given its central position in Earth's water and biogeochemical cycles, an accurate model is critical to water resources (Klemeš, 1988; Arora, 2002) and models of climate (Olesen, 2007; Friend, 2010) as well.

2. The Percolation Model

Soil is a highly heterogeneous medium in which the connectivity of its pores is described by the sample-spanning percolation cluster (SSPC) near the percolation threshold (Katz and Thompson, 1986; Sahimi, 2011, 2023). The SSCP is made of the backbone - the multiply-connected cluster of pores through which flow and transport occur - and the singly-connected or dead-end pores. The SSCP is a self-similar fractal object with a fractal dimension $D_f \approx 1.9$ and 2.5 in two and three dimensions (2D and 3D), as is the backbone with a fractal dimension, $D_b \approx 1.64$ and 1.87 in 2D and 3D, respectively (Sheppard et al., 1999). Likewise, plants' roots system also possesses a fractal structure (Dannowski, 2005; Yang et al., 2022). Although the fractal dimension D_r of plant root depends on the soil depth, its typical value is close to that of the SSPC (Levang-Brilz and Biondini, 2002). Thus, we assume that, $D_r \approx D_f$.

Hunt (2017) employed Rosezweig's ideas to a model of root mass based on percolation theory, in order to develop the scaling relation, $N_p \propto E^{D_f}$, for the NPP. Briefly, since the root radial extent equals transpiration after one growing season and the cumulative transpiration after multiple seasons, and given that the plant's root structure is fractal, its mass equals transpiration raised to the power of fractal dimension D_f . We replace E with its dependence on E_p and P. Since we assume that plants' root network represents a 2D percolation cluster with $D_r \simeq D_f = 91/48 \approx$ 1.9, we obtain, $N_p \propto E_p^{1.9}$. Although the general dependence of N_p on E_p has been verified, the proportionality factor is unknown and, thus, open to speculation.

Chemical weathering - disintegration of soil by chemical reactions - is limited by the rate of transport of cations (solute) in soil, while soil formation is limited by chemical weathering since plants and vegetables cannot use bare rock to grow, but need the cations in the minerals that are transported by flowing water, so that, in the absence of denudation, the chemical weathering depth is equal to the distance that a solute is transported. Solute transport in soil is described by the concepts of percolation theory (Sahimi, 2012; Ghanbarian-Alavijeh et al., 2012), which has been confirmed by experiments (Sahimi et al., 1986; Gist et al., 1990; Ghanbarian-Alavijeh et al., 2012), due to the predominance of flow at or near the percolation threshold. In the percolation representation of solute transport in soil, the total distance x that

a solute is transported by the flowing water is given by

$$x = d_0 \left(\frac{t}{t_0}\right)^{1/D_b} \,. \tag{1}$$

In Eq. (1) d_0 is a characteristic median particle size, and x_0/t_0 is the annual mean flow rate in the unsaturated zone, which has been verified for time scales ranging from a decade to millions of years (Hunt and Ghanbarian, 2016), and over four orders of magnitude of climate variability, representing a rainfall gradient from the Atacama desert to the New Zealand rainforest. Steadystate soil depths is reached when the soil production, defined as dx/dt, is equal to the denudation rate, q_d . The result provides the scaling of x with (P - E) (Hunt et al., 2020, 2021),

$$x = d_0 \left(\frac{P-E}{D_b \phi q_d}\right)^{1/(D_b-1)} , \qquad (2)$$

where ϕ is the soil porosity. The accuracy of Eq. (2) has also been verified over two orders of magnitude variations in precipitation and almost three orders of magnitude variations in the denudation rate q_d .

Although, as described above, approximating transpiration by E results in the scaling relation, $N_p \propto E^{D_f}$, the soil depth is important as well, since carbon is sequestered throughout the soil, and plants' roots extend vertically into the critical zone with a depth roughly equal to the soil depth. Thus, the NPP is proportional to the product of E^{D_f} and x, i.e.,

$$N_p \propto E^{D_f} (P - E)^{1/(D_b - 1)}$$
 (3)

Optimizing N_p with respect to the hydrologic partitioning is accomplished by setting $dN_p/dE = 0$, which results in

$$E = \frac{D_f(D_b - 1)}{1 + D_f(D_b - 1)} P \equiv \alpha(d) P .$$
(4)

where d is the spatial dimension of the system. Since a solute carrying nutrients is transported by water through the backbone of the 3D SSPC, thus one may set $D_b \approx 1.87$. On the other hand, as discussed above, to access soil nutrients carried along by flowing water, trees' roots tend to grow horizontally, i.e., they constitute a 2D SSPC and, therefore, $D_f = 91/48 \approx 1.9$. Hence, we obtain, E = 0.623P, implying that the difference (P - E) is simply proportional to P. The denudation rate q_d is also generally proportional to P, implying that the net result for the soil depth is highly insensitive to climate details and explaining the consistency of soil depths of about 1 m around the world. It, however, adds no significant dependence of N_p on climate variables, so that later substitution of the results for E into the equation for N_p is relevant only for the first factor.

Equation (4) was derived by neglecting both energy and water limitations on the vegetation productivity, when P and E_p are unequal. The energy limitation is taken into account by assuming that only $P = E_p$ is optimized, and that the remaining $(P - E_p)$ runs off, i.e.,

$$E = \alpha(d)E_p . (5)$$

Thus, with $D_f \approx 1.9$ and $D_b \approx 1.87$, we obtain, $E = 0.623E_p$. Water limitation is taken into account by assuming that only the surface covered by vegetation, shown to be proportional to P/E_p by Yang et al. (2009), and assumed here to equal P/E_p , is subject to the optimization, with E = P over the remaining area. The assumption, which neglects known routing of surface water towards vegetation, results in

$$E = P - (1 - \alpha) \left(\frac{P^2}{E_p}\right) = P - 0.377 \left(\frac{P^2}{E_p}\right) .$$
(6)

We emphasize that the numerical factor $0.377 = 1 - 0.623 = 1 - \alpha(d)$, is not a fitted number, but, as derived above, is derived based on percolation theory. Since both D_f and D_b represent universal fractal dimensions, the numerical factor 0.377 = 1 - 0.623 is also universal, which our theory predicts to be the fraction of P that represents E for a 2D plant root system with neither energy nor water limitations, a prediction that is within 1.5% of the measured global mean value of E/P (Hunt et al., 2021).

3. Universal Scaling of Net Primary Productivity

We now show that N_p follows a *universal* scaling law. To begin with, we note that according to Eq. (4), E is proportional to P. Moreover, q_d , the denudation rate, is also proportional to P (Reiners et al., 2003). Thus, in Eq. (2) $(P - E)/q_d \sim \mathcal{O}(1)$ and, as a result, $N_p \propto E^{D_f}$. We define the aridity index as $A_I = E_p/P$. For $A_I < 1$, $E = E_p$ and thus, $N_p \propto E_p^{D_f}$. For $A_I > 1$ we substitute Eq. (6) into $N_p \propto E^{D_f}$ and derive

$$N_p \propto \begin{cases} \alpha(d)^{D_f} E_p^{D_f} & A_I < 1 ,\\ E_p^{D_f} \left[\frac{1}{A_I} - (1 - \alpha) \frac{1}{A_I^2} \right]^{D_f} & A_I > 1 \end{cases}$$
(7)

Thus, Eq. (7) implies that for $A_I < 1$ the ratio $N_p/E_p^{D_f}$ should saturate at a universal value, $\alpha^{D_f} = 0.623^{1.9} \approx 0.4$, whereas for $A_I > 1$ it should be a universal function of the aridity index A_I . This means that separate curves of data taken at various solar irradiance, the flux of radiant energy Rn, should collapse onto a single universal function of A_I . Thus, one has a theory that satisfies simultaneously the criteria of Rosenzweig (1968), namely, that the principal input to NPP should be evapotranspiration, and of Budyko (1958) according to which the variables of relevance to NPP are A_I and E_p , while demonstrating that the critical variable missing from Rosenzweig's analysis, i.e., E_p , was already anticipated by Budyko (1958). The data that Budyko (1974) reported were a function of A_I for various values of the irradiance Rn, which we use as a proxy for potential evapotranspiration E_p .

4. Comparison with Experimental Data

Beyond the difficulties in prediction lie those in finding suitable data to test the predictions of the percolation model, as very few researchers have chosen to represent N_p as a function of $A_I = E_p/P$. As mentioned above, one of the few who made such a systematic study was Budyko (1974). We have used a digitized version of his original data and figures, reported by Gupta et al. (2002), for evaluating the accuracy of the predicted universal scaling of N_p . Given that, for example, Adams et al. (2004) needed 20 journal pages to describe ten representative models of NPP and 15 more to list the equations and their constants, we believe that there is an important place for a model with only two parameters, P and E_p , which are simultaneously independent variables in the related fundamental problem of hydrology, namely, the water balance.

Figure 1 presents the original data of Budyko for NPP versus P/E_p . Figure 1(b) exhibits the data, rescaled according to Eq. (7). Interestingly, the crossover in the data from energy- to water-limited regimes occurs at $A_I < 1$ around 0.8. We, therefore, present simultaneously the predictions of Eq. (7), but scaled to a crossover from energy to water limitations at $A_I/0.8 = 1$. As a consequence of the tendency for solar energy to arrive continuously, but for water to arrive in lumps, ecosystems are water-limited even for A_I less than, but close to 1. One can see that the simple uncertainty in the position of the crossover point from water- to energy-limitations suffices to map out the range of the data in arid ecosystems, since the predicted curve maps out the lower boundary of the data in both regimes.

In Figure 2, we compare the individual sets of data with the predictions, but using the modification of Eq. (7) scaled to $A_I/0.8$. As Figure 2 demonstrates, the predicted N_p are around the 1:1 line, indicating accurate theoretical predictions by the percolation model.

Discussion

In ecology, null models (Harvey et al., 1983) can be developed to determine whether data are "structured." In physics, scaling plots of data reveal universality of certain functions. The scaling plot of Figure 2 reveals that, even in the presence of some discrepancies, the fundamental structure of the data is compatible with the predictions of Eq. (7). The inputs are climatic variables P and E_p . The origin of the structure in the ecological dataset of Budyko (1974) can be thought of as the hypothesis that ecosystems, or clades (Doolittle, 2017), are selected on the basis of productivity within limitations primarily of energy and water, in such a way that the ensemble of species fills the niches that are most productive. Thus, given a set of inputs, the plant ecosystem that we actually find is the one that is best adapted to growing and reproducing itself, subject to the climate limitations present.

Other limiting factors may exist: there are portions of Figure 1 where the curves fall short of the predicted productivity. Approximations made in order to develop theory lead to differences with the data. For example, we underestimate the productivity of ecosystems with $A_I < 1$ and small values of the solar irradiance Rn. We, however, have based our predictions on the concept that total biomass scales consistently with below-ground biomass. Thus, it is possible that in such ecosystems where water and nutrients are not as important limiting factors as light, a larger share of plant biomass is found above-ground than on average across the remaining ecosystems, leading to the underprediction. In order to realize the full predictive potential of the theory, we also have to admit the potential relevance of water limitations to systems in which the input energy, as measured in potential evapotranspiration, is less than the total precipitation. The reason for such an asymmetry in the response of vegetation to the water balance - a crossover from energy- to water-limitations before the energy can evaporate all the water - presumably lies in the distinct characters of solar energy and precipitation inputs to the ecosystem: Water arrives less regularly and in lumps, some of it flowing off when not needed and unavailable when needed, whereas energy arrives nearly continuously.

In a related study, Huxman et al. (2004) found convergence to a common water-use strategy in the arid limit. Our results appear to be suggest a similar convergence. We note also that our scaling analysis of NPP as a function of aridity index in Figure 2 illuminates a striking tendency of ecosystems towards common behavior in the arid limit, in accord with the discussion of Huxman et al. (2004).

5. Summary

In a recent theory for the water balance, the fraction of precipitation P returned to the atmosphere through evapotranspiration E was determined by optimizing the net primary productivity, NPP, of ecosystems with respect to the principal hydrologic fluxes, E and Q, the run-off. Here we substituted this result for E as a function of precipitation and potential evapotranspiration into an existing result for the NPP of ecosystems as a function of E. The result gives NPP as a universal function NPP(P, Ep) with a single unknown constant, which plays the role is of a universal carbon assimilation efficiency. Comparison with a compilation of the known global data (at the time) by Budyko in his book, Climate and Life (1974) demonstrates a strong predictive capability of the derived relationship with an R^2 of 0.97 across biomes and climate variables. The predicted dependence on aridity index is strong in arid regimes, but saturates at low aridity. In contrast to the data, the predicted form of this saturation involves a discontinuous change in slope near aridity index 1. Although this introduces a slight discrepancy in the comparison with observation for NPP, the relevance of the slope discontinuity is proven in (separate) comparisons of streamflow elasticity with observation.

Open Research

Data used in this study (A.G. Hunt et al., 2023) are available at:

http://www.hydroshare.org/resource/eee62676c9874dc4b3bc44712c36315a as well as in the Supplemental Material section.

References

- Adams, B., White, A., & Lenton, T.M. (2004). An analysis of some diverse approaches to modelling terrestrial net primary productivity. *Ecol. Model.*, 177(3-4), 353.
- Anderson, S.P., von Blanckenburg, F., & White, A.F. (2007). Physical and chemical controls on the critical zone. *Elements*, 3(5), 315.
- Arora, V. (2002). Modeling vegetation as a dynamic component in soil-vegetation-atmosphere transfer schemes and hydrological models. *Rev. Geophys.*, 40(2) 3.

- Brantley, S.L., Goldhaber, M.B., & Ragnarsdottir, K.V. (2007). Crossing disciplines and scales to understand the critical zone. *Elements*, 3(5), 307.
- Budyko, M.I. (1958). The heat balance of the Earth's surface. Translated by Stepanova, N.A., US Department of Commerce, Weather Bureau.
- Budyko, M.I. (1974). *Climate and Life* (Academic Press, New York).
- Cramer, W., et al. (1999). Comparing global models of terrestrial net primary productivity (NPP): overview and key results. *Global Change Biol.*, 5(S1), 1.
- Dannowski, M. (2005). Fractal geometry and root system structures of heterogeneous plant communities. *Plants and Soil*, 272, 61.
- de Wit, C.T. (1965). Photosynthesis of leaf canopies, Agricultural Research Report no. 663 PUDOC Wageningen.
- D'Dodorico, P., et al. (2018). The global food-energy-water nexus. Rev. Geophys., 56, 456.
- Falkenmark, M., & Rockström, J. (2006). The new blue and green water paradigm: Breaking new ground for water resources planning and management. J. Water Resour. Planning Management, 132(3), 129.
- Friend, A.D. (2010). Terrestrial plant production and climate change. J. Exp. Botany, 61(5), 1293.
- Gentine, P., D'Dodorico, P., Linter, B.R., Sivandran, G., & Salvucci, G. (2012). Interdependence of climate, soil, and vegetation as constrained by the Budyko curve. *Geophys. Res. Lett.*, 39, L19404.
- Ghanbarian-Alavijeh, B., Skinner, T.W., & Hunt, A.G. (2012). Saturation dependence of dispersion in porous media. *Phys. Rev. E*, 86, 066316.
- Gist, G.A., Thompson, A.H., Katz, A.J., & Higgins, R.L. (1990). Hydrodynamic dispersion and pore geometry in consolidated rock. *Phys. Fluids A*, 2, 1533.

- Gupta, V.K., Milne, B.T., & Poveda, G. (2002). Towards a scale invariant evapotranspiration equation using real-space renormalization. *Eos Trans. AGU*, 83(47), Fall Meet. Suppl., Abstract H62F-08.
- Harvey, P.H., Colwell, R.K., Silvertown, J.W., & May, R.M. (1983). Null models in ecology. Annual Rev. Ecol. Systematics, 14, 189.
- Hillel, D. (2003). *Environmental Soil Physics* (Elsevier, Amsterdam).
- Horton, R. (1931). The field, scope, and status of the science of hydrology. Trans. Am. Geograph. Union, 12, 189.
- Huang, Y., Li, Y.P., Chen, X., & Ma, Y.G. (2012). Optimization of the irrigation water resources for agricultural sustainability in Tarim River Basin, China. Agricul. Water Management, 107, 74.
- Hunt, A.G. (2017). Spatio-temporal scaling of vegetation growth and soil formation: Explicit predictions. Vadose Zone J., 16, 1.
- Hunt, A.G., Faybishenko, B.A., & Ghanbarian, B. (2021). Predicting characteristics of the water cycle from scaling relationships. *Water Resour. Res.*, 57, e2021WR030808.
- Hunt, A.G., Faybishenko, B.A., & Powell, T.L. (2020). A new phenomenological model to describe root-soil interactions based on percolation theory. *Ecolog. Modelling*, 433, 109205.
- Hunt, A.G., & Ghanbarian, B. (2016). Percolation theory for solute transport in porous media: Geochemistry, geomorphology, and carbon cycling. Water Resour. Res., 52(9), 7444.
- Hunt, A. G., M. Sahimi, B. Ghanbarian (2023). Net Primary Productivity, HydroShare, http://www.hydroshare.org/resource/eee62676c9874dc4b3bc44712c36315a
- Huxman, T.E., et al. (2004). Convergence across biomes to a common rain-use efficiency. Nature, 429, 651-654.
- Katz, A.J., & Thompson, A.H. (1986). Quantitative prediction of permeability in porous rock. *Phys. Rev. B*, 34, 8179 (1986).
- Klemeš, V. (1988). A hydrological perspective. J. Hydrol., 100(1-3), 3-28.

- Krause-Jensen, D., & Sand-Jensen, K. (1998). Light attenuation and photosynthesis of aquatic plant communities. *Limnology and Oceanography*, 43, 396.
- Krishnaswamy, J., Bawa, K.S., Ganeshaiah, K.N., & Kiran, M. C. (2009). Quantifying and mapping biodiversity and ecosystem services: Utility of a multi-season NDVI based Mahalanobis distance surrogate. *Remote Sensing of Environment*, 113(4), 857.
- Levang-Brilz, N., & Biondini, M.E. (2002). Growth rate, root development and nutrient uptake of 55 plant species from the Great Plains Grasslands, USA. *Plant Ecol.*, 165, 117.
- Maxwell, R.M., & Condon, L.E. (2016). Connections between groundwater flow and transpiration partitioning. *Science*, 353(6297), 377.
- National Academy of Sciences (1991). Opportunities in the Hydrologic Sciences, https:// nap.nationalacademies.org/catalog/1543/opportunities-in-the-hydrologic-sciences
- Oehri, J., Schmid, B., Schaepman-Strub, G., & Niklaus, P. A. (2017). Biodiversity promotes primary productivity and growing season lengthening at the landscape scale. *Proc. Natl* Acad. Sci, 114(38), 10160.
- Olesen, J.E. (2007). Uncertainties in projected impacts of climate change on European agriculture and terrestrial ecosystems based on scenarios from regional climate models. *Climatic Change*, 81(1), 123.
- Raich, J.W., et al. (1991). Potential net primary productivity in South America: application of a global model. *Ecol. Applications*, 1(4), 399.
- Rosenzweig, M.L. (1968). Net primary productivity of terrestrial communities: prediction from climatological data. Am. Nat., 102, 67.
- Sahimi, M. (2011). Flow and Transport in Porous Media and Fractured Rock, 2nd ed (Wiley-VCH, Weinheim).
- Sahimi, M. (2012). Dispersion in porous media, continuous-time random walks, and percolation. Phys. Rev. E, 85, 016316.
- Sahimi, M. (2023). Applications of Perclation Theory, 2nd ed (Springer, New York).

- Sahimi, M., Hughes, B.D., Scriven, L.E., & Davis, H.T. (1986). Dispersion in flow through porous media: I. One-phase flow. *Chem. Eng. Sci.*, 41, 2103.
- Schlesinger, W.H., & Jasechko, S. (2014). Transpiration in the global water cycle. Agric. For. Meteorol., 189, 115.
- Sheppard, A.P., Knackstedt, M.A., Pinczewski, W.V., & Sahimi, M. (1999). Invasion percolation: new algorithms and universality classes, J. Phys. A, 32, L521.
- Tapley, B.D., et al. (2019). Contributions of GRACE to understanding climate change. Nat. Clim. Chang., 5(5), 358.
- Webb, W.L., Lauenroth, W.K., Szarek, S.R., & Kinerson, R.S. (1986). Primary production and abiotic controls in forests, grasslands, and desert ecosystems of the United States. *Ecology*, 64, 134.
- Yang, D., et al. (2009). Impact of vegetation coverage on regional water balance in the nonhumid regions of China. *Water Resour. Res.*, 45, W00A14.
- Yang, Q., Cheng, W., Hao, Z., Zhang, Q., Yang, D., Teng, D., Zhang, Y., Wang, X., Shen, H., & Lei, S. (2022). Study on the fractal characteristics of the plant root system and its relationship with soil strength in tailing ponds. Wireless Commun. Mobile Comput., 2022, Article 9499465.



Figure 1: Comparison of the individual predictions for the data of Budyko (1974). In this case, the curve representing the prediction for Rn = 70 is adjusted in order to produce the best match with the data at both large and small $A_I > 1$, and all the other predictions are scaled by the ratio of Rn/70, raised to the 1.9 power (the fractal dimension of two-dimensional percolation cluster). Owing to its superior performance, however, the scaling variable was chosen as $A_I/0.80$.



Figure 2: Comparison of the scaled plot of the data of Budyko (1974) with one of the predicted curves using Eq. (7), as well as its modification by substituting $A_I/0.80$ for A_I .